Author’s response to reviews

Title: Improving risk prediction accuracy for new soldiers in the U.S. Army by adding self-report survey data to administrative data

Authors:
Samantha Bernecker (bernecker_samantha@fas.harvard.edu)
Anthony Rosellini (ajrosell@bu.edu)
Matthew Nock (nock@wjh.harvard.edu)
Wai Tat Chiu (echiu@hcp.med.harvard.edu)
Peter Gutierrez (peter.gutierrez@va.gov)
Irving Hwang (hwang@hcp.med.harvard.edu)
Thomas Joiner (joiner@psy.fsu.edu)
James Naifeh (james.naifeh.ctr@usuhs.edu)
Nancy Sampson (sampson@hcp.med.harvard.edu)
Alan Zaslavsky (zaslavsk@hcp.med.harvard.edu)
Murray Stein (mstein@ucsd.edu)
Robert Ursano (robert.ursano@usuhs.edu)
Ronald Kessler (Kessler@hcp.med.harvard.edu)

Version: 2 Date: 20 Feb 2018

Author’s response to reviews:

N/A - we were informed that a response to reviewers is no longer needed at this stage of the editorial process.